Author’s response to reviews

Title: Phoyunnanin E inhibits migration of non-small cell lung cancer cells via suppression of epithelial-to-mesenchymal transition and integrin αv and integrin β3

Authors:

NAREERAT PETPIROON (petpiroon_bright@hotmail.com)

BOONCHOO SRITULARAK (pithi.c@chula.ac.th)

Pithi Chanvorachote (pithi_chan@yahoo.com)

Version: 3 Date: 05 Dec 2017

Author's response to reviews:

Response to reviewer

Please make the following revisions to your manuscript:

1. Please revise your competing interest statement to account for all possible competing interests, not just financial ones.

Response: We have changed it to “The authors declare that they have no competing interests.”

2. Thank you for including a data availability statement. This statement should detail the location of the raw data underlying the conclusions made in this study. We note that only summarized data is presented in this manuscript, so we would like to ask that you amend this statement. If there are any ethics restrictions preventing the sharing of the raw data, please state this. If you do not wish to share the data, please include a reason why it will not be shared in this section.

Response: we changed to “The datasets supporting the conclusions of this article are included within the article. The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.”